Author's response to reviews

Title: Choledochoduodenal fistula and pneumobilia secondary to gallbladder cancer

Authors:

Elham Dadzan (elham_d@hotmail.com)
Hossein Akhondi Asl (h68akhond@hotmail.com)

Version: 2 Date: 13 September 2011

Author's response to reviews:

Hello,

This manuscript was initially send to BMC Gastroenterology but it was thought that it might be more of general internal medicine interest. The case in fact was a diagnostic dilemma and it took several days to reach a final diagnosis. During these days multiple complications also happened.

We thought this might be of interest for people who are working up similar cases.

Thanks much